Craig Chambers CSE 505 (Fall ’95) Homework Assignments

Homework Assignment #4 Due Friday 10/27

From Chapter 3 exercises: 1.3a 2.2: also explain the general principle involved, for an eagerly-evaluated parameter of any syntactic domain.

2.6: also explain why it might be much more difficult to implement eagerly-evaluated abstractions with parameters than without, at least for expression abstractions.

3.2b: also first give an example program that would lead to that substitution being applied when simplifying by the copy rule.

4.3a: you do not need to define the semantics of eagerly- and lazily-evaluated location parameters; this is done ad nauseum in the text.

4.4c: by “explain the semantics” apply the copy rule to simplify the programs and/or use the semantic definitions of the constructs to compute the meaning (in the technical sense) of the programs. 5.2 7.1de: you may use a general version of abstractions and parameters, written in terms of U and V, etc., rather than replicating the same idea over and over. 7.2a